Author's response to reviews

Title: International Medical Graduates (IMGs) Needs Assessment Study: Comparison Between Current IMG Trainees and Program Directors

Authors:

Rachelle Zulla (rachel.zulla@utoronto.ca)
Mark Otto Baerlocher (mark.baerlocher@utoronto.ca)
Sarita Verma (sarita.verma@utoronto.ca)

Version: 6 Date: 12 August 2008

Author's response to reviews:

As requested by editor, the following revisions have been made:

1) 'Submitted on April 1, 2008 to BMC Medical Education Accepted, in principle, May 12, 2008' has been removed from the title page

2) Sentence case has been used in the title

3) indicated in writing that all authors have reviewed and approved this manuscript

4) table shading has been eliminated; main headings are in caps